Author's response to reviews

Title: Oxaliplatin-based versus irinotecan-based hyperthermic intraperitoneal chemotherapy (HIPEC) in patients with peritoneal metastasis from appendiceal and colorectal cancer: a retrospective analysis

Authors:

Gabriel Glockzin (gabriel.glockzin@ukr.de)
Michael Gerken (michael.gerken@ukr.de)
Sven A Lang (sven.lang@ukr.de)
Monika Klinkhammer-Schalke (monika.klinkhammer-schalke@ukr.de)
Pompiliu Piso (pompiliu.piso@barmherzige-regensburg.de)
Hans J Schlitt (hans.schlitt@ukr.de)

Version: 4 Date: 22 September 2014

Author's response to reviews: see over
COVER LETTER

MS: 2044003857127794
Title: Oxaliplatin-based versus irinotecan-based hyperthermic intraperitoneal chemotherapy (HIPEC) in patients with colorectal peritoneal metastasis: a retrospective analysis
Authors: Gabriel Glockzin, Michael Gerken, Sven A. Lang, Monika Klinkhammer-Schalke, Pompiliu Piso and Hans J. Schlitt

Dear editorial board,

with this cover letter you will find the revised manuscript and a point-by-point response to the editorial request.

Best regards
Gabriel Glockzin

EDITORIAL REQUEST:

#1 New title as the current one is misleading given the mixed histology (no just colorectal) assessed in this manuscript.

The title has been changed to ‘Oxaliplatin-based versus irinotecan-based hyperthermic intraperitoneal chemotherapy (HIPEC) in patients with peritoneal metastasis from appendiceal and colorectal cancer: a retrospective analysis’.

#2 Quick word/grammar review as there were couple repeated words noted.

A word/grammar review has been performed. The repeated words have been corrected/removed in/from the revised manuscript.

To conform to the journal style the figures have been removed from the revised manuscript and are provided as separate files.